Author’s response to reviews

Title: The association between dietary isoflavones intake and gastric cancer risk: a meta-analysis of epidemiological studies

Authors:
Jie You (zzu_youjie@126.com)
Yafei Sun (sunyafei66@126.com)
Yacong Bo (boyacong@163.com)
Yiwei Zhu (yiwei0726@163.com)
Dandan Duan (15236273181@163.com)
Han Cui (zhuxicl@126.com)
Quanjun Lu (Lqjnutr@zzu.edu.cn)

Version: 3 Date: 05 Apr 2018

Author’s response to reviews:

Dear editor,

Thank you for giving us another chance to resubmit our manuscript "The association between dietary isoflavones intake and gastric cancer risk: a meta-analysis of epidemiological studies" (PUBH-D-17-03047). We have revised the paper according to your comments.

The following is our point-by-point responses to editor’s comments.

1. Response to comment: The declaration section is now missing both the 'Availability of Data and Materials' and 'Ethics and Consent to Participate' sections. Please can you included these sections, both of which can say 'Not Applicable'.

Response: Thanks for your valuable advices. We have added the two missing sections, according to your suggestions.

2. Response to comment: Please can you include institutional email addresses for all authors on the Title Page. The corresponding author should still be indicated.
Response: Thanks for your kind suggestions. We have added the institutional email addresses for all authors on the Title Page and the corresponding author has been indicated.

Special thanks to you for your good comments. Best wishes!

Thank you and all the reviewers for the kind advices.

Sincerely yours,

Quanjun Lu